\chapter{Synopsis of Distribution Theory}

\blockcquote{Strichartz1994}{%
  \textins{Distribution theory} can be thought of as the completion of differential calculus,
  just as the other great revolution, measure theory \textelp{},
  can be thought of as the completion of integral calculus.
}
Just more text.

\section{Operator Valued Distributions}

For a historical perspective see
\citetitle{Wightman1996}
by
\textcite{Wightman1996}.

\nocite{Halperin1952}
\nocite{Lighthill1958}
\nocite{Gårding1959}
\nocite{Schwartz1966}
\nocite{Trèves1967}
\nocite{Jones1982}
\nocite{Richards1990}
\nocite{Friedlander1999}
\nocite{Hoskins2005}
\nocite{Hoskins2009}


\chapterbib
\cleardoublepage
